

**Name of Journal:** *World Journal of Clinical Cases*

**Number ID:** 01109127

**Manuscript Type:** Review

**Title:** Kikuchi-Fujimoto disease: a narrative review

**Short Title:** Kikuchi-Fujimoto disease

**Title Submit Date:** 2022-09-22

**Title Accepted Date:** 2022-09-29

**Manuscript Submission Deadline:** 2022-12-28

To  
The Editor-in-Chief  
*World Journal of Clinical Cases*

Sir,

The revised manuscript titled “Kikuchi-Fujimoto disease: a narrative review” is being submitted to your esteemed journal. The manuscript has been revised to include answers to the reviewers’/editorial comments.

#### **Peer-review report**

**Reviewer #1:** However the review topic is interesting there are several comments and recommendations: - **Thanks for evaluating our manuscript.**

**Comment-1:** Abstract looks like an introduction rather than a summary including background, methods and conclusion of your search.

**Answer-1:** This was not a systematic review and the abstract format was as per instructions for the authors. The title has been revised to “Kikuchi-Fujimoto disease: a comprehensive review” to reflect an extensive review of the topic.

**C-2:** SLE is a severe and an important complication of KFD, you discussed it well however update is required ( see as example, Tarabichi S, et al. Kikuchi-Fujimoto disease and systemic lupus erythematosus—an enigmatic association for clinicians and pathologists: Case report and short literature review. *Romanian Journal of Rhinology*. 2022 Jan;12(45):11-21.

**A-2:** Agreed. A para (page 19), and the suggested reference citation (Ref 167) has been added under the subheading “KFD and Systemic lupus erythematosus”

**C-3:** Also you miss some cases reported for SARS-cov vaccine related KFD

**A-3:** Reference citation updated to include more references (ref 35-48) relevant to the topic avoiding duplication of information.

**C-4:** You provide a title of (comment) at the end of the article which is vague, and if you can conclude your unique information and conclude your recommendation would be better.

**A-2:** Modified as suggested

**Reviewer #2:**

**C-1:** Please do a grammar check again. You can use Grammarly to correct the grammar and spelling

**A-1:** Ok, done

Hope you find the manuscript suitably revised.

Sincerely,

**Dr. Vikram K Mahajan**

Department of Dermatology, Venereology & Leprosy

Dr. Radhakrishnan Government Medical College

Hamirpur 177001

Himachal Pradesh (India)